Author's response to reviews

Title: Factors associated with recognition and prioritization for falling, and the effect on fall incidence in community dwelling older adults

Authors:

Sofie Jansen (s.jansen@amc.uva.nl)
Jolanda Schoe (jolanda.schoe@gmail.com)
Marjon van Rijn (m.vanrijn@amc.uva.nl)
Ameen Abu-Hanna (a.abu-hanna@amc.uva.nl)
Eric Moll van Charante (e.p.mollvancharante@amc.uva.nl)
Nathalie van der Velde (n.vandervelde@amc.uva.nl)
Sophia de Rooij (s.e.derooij@amc.uva.nl)

Version: 5
Date: 6 December 2015

Author's response to reviews:

Amsterdam, December 6, 2015

Dear Dr. Akai,

Many thanks for considering our paper for publication. Below, we have addressed the final comments. Attached find the revised manuscript.

Yours sincerely,

Sofie Jansen, PhD candidate
Dept. of Medicine, section of Geriatric Medicine
Academic Medical Center, University of Amsterdam
Meibergdreef 9, room F4-218
1105AZ Amsterdam, the Netherlands
s.jansen@amc.uva.nl

Comments:

1. Format the Abstract into four sections: Background, Methods, Results and Conclusions.

   We have formatted the abstract in four sections.

2. Include the TRN (Trial Registration Number) and date of registration.

   The TRN and date of registration were added.

3. Conclusions section.
We have specified which section was the conclusion section (page 12)